Author’s response to reviews

Title: Prevalence and antimicrobial susceptibility of asymptomatic bacteriuria among women with pelvic organ prolapse in Abakaliki, South-East Nigeria.

Authors:
Kenneth Ekwedigwe (ekwedigwe@yahoo.com)
Ileogben Sunday-Adeoye (juladeoye@yahoo.com)
Monday Eliboh (osazemonday@gmail.com)
MARADONA ISIKHUEMEN (maradona4real2002@yahoo.com)
Henry Uro-Chukwu (henryuchukwu@yahoo.co.uk)
Paul Ezeonu (olisaemekapaulo@yahoo.com)
Babafemi Daniyan (babafemidaniyan@yahoo.com)
Emmanuel Yakubu (emayaks@yahoo.com)

Version: 4 Date: 15 Mar 2018

Author’s response to reviews:

The amount of overlap in your paper is still not acceptable for publication. I have attached a document showing the overlap - please revise these sections as necessary.

Response: Sentences in the introduction and methods section have been altered to avoid overlap. Thank you